Author's response to reviews

Title: '20 days protected learning' - students' experiences of an Overseas Nurses Programme - 4 years on: A retrospective survey

Authors:

Gill Jordan (gjordan@bournemouth.ac.uk)
Petra Brown (pbrown@bournemouth.ac.uk)

Version: 4 Date: 13 April 2011

Author's response to reviews:

Page 4: Figure 1 inserted into text
Page 1: UK added to authors' address & email addresses added
Tracking and highlighting removed throughout text
Typos corrected